Author's response to reviews

Title: Role of Culex and Anopheles mosquito species as potential vectors of rift valley fever virus in Sudan outbreak, 2007

Authors:

   AlaaEddeen M Seufi (alaaseufi@yahoo.com)
   Fatma H Galal (fatmahgalal@yahoo.com)

Version: 9 Date: 2 March 2010

Author's response to reviews:

one typographical error was corrected